Author's response to reviews

Title: Screening of Depression In Primary Care Setting - Prevalence and Instrument

Authors:

Waleed Al Qadhi (dwaqs@hotmail.com)
Saeed Ur Rahman (rahmans@ngha.med.sa)
Mazen Saleh Ferwana (ferwanam@ngha.med.sa)
Abdulmajeed Imad Addin (abdulmajeedim@ngha.med.sa)

Version: 2 Date: 17 September 2013

Author's response to reviews: see over
17 September 2013

The Editor
BMC Psychiatry

Dear Sir:

We confirm that this manuscript “Adult Depression Screening in Saudi Primary Care: Prevalence, Instrument and Cost” has not been published elsewhere and is not under consideration by another journal.

The authors have approved the manuscript and agree with its submission to BMC Psychiatry Journal.

Thank you.

Sincerely,

MAZEN SALEH FERWANA, MD, ABFM, JBFM, PhD
Department of Family Medicine and Primary Healthcare
King Abdulaziz Medical City, National Guard Health Affairs
P.O. Box 22490, Riyadh 11426
Kingdom of Saudi Arabia
Phone: (966) 11-4299999 ext 91167/91159
Fax: (966) 11-4299999 ext 91193
Email: drmazen99@yahoo.com; ferwanam@ngha.med.sa
Adult Depression Screening in Saudi Primary Care: Prevalence, Instrument and Cost

Al-Qadhi WA¹, Saeed Ur Rahman², Ferwana MS³, Al ⁴

¹R4 Family Medicine Resident,

²Consultant, Associate DED, Community Medicine, Department of Family Medicine and Primary Health Care,

³Consultant, Associate Professor, Family Medicine and Co-Director of National & Gulf Center for Evidence Based Health Practice, King Abdulaziz Medical City, Riyadh, Saudi Arabia

⁴Public Health Specialist, Staff Physician, Department of Family and Community Medicine, King Abdulaziz Medical City- Riyadh

Corresponding Author:

MAZEN SALEH FERWANA, MD, ABFM, JBFM, PhD

Department of Family Medicine and Primary Healthcare

King Abdulaziz Medical City, National Guard Health Affairs

P.O. Box 22490, Riyadh 11426

Kingdom of Saudi Arabia

Phone : (966) 11-4299999 ext 91167/91159

Fax : (966) 11-4299999 ext 91193

Email : drmazen99@yahoo.com; ferwanam@ngha.med.sa